Author’s response to reviews

Title: Isolated Palmar Dislocation of Distal Radioulnar Joint: A New Mechanism of Injury: A case report

Authors:
Xianke Lin (1515038@zju.edu.cn)
Hui Shen (shenhui33@sina.com)
Hui Lu (hitman1982@hotmail.com; huilu@zju.edu.cn)

Version: 1 Date: 23 Jun 2019

Author’s response to reviews:

Dear Prof. Robin L. Cassady-CAin:

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. The response to your comments are listed below this letter.

We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Yours sincerely, Hui Lu M.D.

Department of Hand Surgery, The First Affiliated Hospital, College of Medicine, ZheJiang University. Huilu@zju.edu.cn
Replies to Prof. Ulunay KANATLI, M.D. (Reviewer 1)

Thank you for your hard review and serious modification.

1. The direct impact may resemble fall from height, so the authors need to support this further.

Answer: Thanks, we carefully reviewed our case, the patient had no history of falling from height, and if there are similar patients in future, we will pay special attention to the medical history.

Replies to Reviewer 2

Thank you for your hard review and serious modification.

1. The discussion should be longer. The table is really nice and the authors need to highlight and discuss this more.

Answer: Thanks, we add more details of the diagnosis (see in page 5, line 92~95). We add more details of the treatment (see in page 5, line 105~109)